Author's response to reviews

Title: Determinants of prescribed drug use among pregnant women in Bahir Dar city administration, Northwest Ethiopia: A cross sectional study.

Authors:

    Chanie Admasie (chaneadmasse@gmail.com)
    Belaynew Wassie (bewassie@gmail.com)
    Gedefaw Abeje Fekadu (abejegedefaw@gmail.com)

Version: 6 Date: 13 September 2014

Author's response to reviews:

Date 13 September 2014

To: BMC Editors
Subject: Submitting revised version of our manuscript

Dear Editors, first we would like to thank you for the constructive comments we received.

Based on your comments, we have changed the arrangement of tables and figure legends. According to the instruction for authors, we put the tables after the references. Tables are also divided into fields by black lines. We also put legends of figures after list of tables. We removed legends from figures.

We have also provided point by point response to reviewers’ comments. We are eager to make further corrections if any. We have attached the revised manuscript, point by point response and corrected figures.

Kind regards,

Chanie Admassie
Gedefaw Abeje
Belaynew Wassie